Author's response to reviews

Title: Hazards of tube thoracostomy in patients on a ventilator

Authors:

Kasra Shaikhrezai (kasrash@gmail.com)
Vipin Zamvar (zamvarv@hotmail.com)

Version: 4 Date: 18 March 2011

Author's response to reviews:

Dear Sir / Madam

The only change which is a typographic mistake is to omit the first bolded word "Injury" in “References” section as it has been repeated twice: reference number 5, line 2. The rest of the manuscript is absolutely fine and ready for publication.

Kind regards
Shaikhrezai et al